

Citation: Moore SM, Oidtman RJ, Soda KJ, Siraj AS, Reiner RC, Jr., Barker CM, et al. (2020) Leveraging multiple data types to estimate the size of the Zika epidemic in the Americas. PLoS Negl Trop Dis 14(9): e0008640. https://doi.org/10.1371/ journal.pntd.0008640

Editor: Elvina Viennet, Australian Red Cross Lifelood, AUSTRALIA

Received: April 14, 2020

Accepted: July 25, 2020

Published: September 28, 2020

Copyright: © 2020 Moore et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All of the cumulative Zika case data used in our analysis are located in a public Github repository: https://github.com/ mooresea/Zika_IAR.

Funding: Funding was provided by NIH supplement Grant R01 Al102939-05 (www.nih. gov) to TAP and RAPID grant DEB 1641130 from the National Science Foundation (www.nsf.gov) to TAP and RCR. TAP was also supported by a DARPA (www.darpa.mil) Young Faculty Award (D16AP00114). The funders had no role in study RESEARCH ARTICLE

Leveraging multiple data types to estimate the size of the Zika epidemic in the Americas

Sean M. Moore^{1*}, Rachel J. Oidtman¹, K. James Soda^{1*}, Amir S. Siraj¹, Robert C. Reiner Jr.², Christopher M. Barker³, T. Alex Perkins^{1*}

1 Department of Biological Sciences and Eck Institute for Global Health, University of Notre Dame, Notre Dame, Indiana, United States of America, 2 Institute for Health Metrics and Evaluation, Seattle, Washington, United States of America, 3 Department of Pathology, Microbiology, and Immunology, School of Veterinary Medicine, University of California, Davis, Davis, California, United States of America

Current address: Department of Mathematics and Statistics, Quinnipiac University, Hamden, Connecticut, United States of America

* smoore15@nd.edu (SMM); taperkins@nd.edu (TAP)

Abstract

Several hundred thousand Zika cases have been reported across the Americas since 2015. Incidence of infection was likely much higher, however, due to a high frequency of asymptomatic infection and other challenges that surveillance systems faced. Using a hierarchical Bayesian model with empirically-informed priors, we leveraged multiple types of Zika case data from 15 countries to estimate subnational reporting probabilities and infection attack rates (IARs). Zika IAR estimates ranged from 0.084 (95% Crl: 0.067–0.096) in Peru to 0.361 (95% Crl: 0.214–0.514) in Ecuador, with significant subnational variability in every country. Totaling infection estimates across these and 33 other countries and territories, our results suggest that 132.3 million (95% Crl: 111.3-170.2 million) people in the Americas had been infected by the end of 2018. These estimates represent the most extensive attempt to determine the size of the Zika epidemic in the Americas, offering a baseline for assessing the risk of future Zika epidemics in this region.

Author summary

During the recent Zika epidemic in the Americas millions of people were likely infected, but the true size of the epidemic is unknown because of gaps in the surveillance system. The infection attack rate (IAR)—defined as the proportion of the population that was infected over the course of the epidemic—has important implications for the longer-term epidemiology of Zika in the region, such as the timing, location, and likelihood of future outbreaks. To estimate the IAR and the total number of people infected, we leveraged multiple types of Zika case data from 15 countries and territories where subnational data were publicly available. Datasets included confirmed and suspected Zika cases in pregnant women and in the total population, Zika-associated Guillan-Barré syndrome cases, and cases of congenital Zika syndrome. We used a hierarchical Bayesian model with empirically-informed priors that leveraged the different case report types to simultaneously estimate national and subnational reporting probabilities, the fraction of symptomatic design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

infections, and subnational IARs. In these 15 countries and territories, estimates of Zika IAR ranged from 0.084 (95% CrI: 0.067–0.096) in Peru to 0.361 (95% CrI: 0.214–0.514) in Ecuador. Totaling these infection estimates across these and 33 other countries and territories in the region, our results suggest that 132.3 million (95% CrI: 111.3-170.2 million) people in the Americas were infected with ZIKV by the end of 2018.

Introduction

Zika virus is a mosquito-borne pathogen that was first identified in Uganda in 1947 [1]. Smaller Zika outbreaks have occurred in Africa, Asia, and the Pacific islands since its discovery, but there had been no confirmed cases in the Americas (excluding Easter Island, Chile) prior to the first confirmation of a Zika case in Brazil in May, 2015 [2]. Subsequent to its discovery in Brazil, the epidemic spread rapidly and cases were reported throughout the Americas over the next two years [3]. The Zika epidemic generated a large amount of concern in the public health community and the general public, leading to a declaration of a Public Health Emergency of International Concern by the World Health Organization (WHO) in February, 2016 [4], because of the discovery of a link between ZIKV infection in pregnant women and congenital Zika syndrome (CZS) in newborns [5–8]. ZIKV infection is also associated with rare but serious neurological disorders, particularly Guillan-Barré syndrome [9]. Following the large epidemic from 2015-2017, substantially fewer Zika cases have been reported in 2018 and 2019 [3].

Now that the initial wave of the ZIKV epidemic in the Americas has passed, there are a number of unanswered questions about what will happen next. If the remaining population at risk is large, then additional outbreaks in the coming years are still possible. On the other hand, modeling has suggested that if a large proportion of the population is now immune, herd immunity will likely prevent another large epidemic for more than a decade assuming life-long immunity following recovery [10]. This scenario could lead to a slow buildup of new susceptible individuals over time, and of particular concern, an eventual buildup of susceptible women of childbearing age if a ZIKV vaccine is not licensed and broadly deployed [10–12]. The number of recent ZIKV infections could also have relevance for the epidemiology of dengue virus (DENV) in the region [10]. There is evidence of an interaction between ZIKV and DENV via the human immune response to infection with either virus [13–17]. If a ZIKV infection provides any temporary cross-protection to DENV, then the reduction in dengue incidence in several Latin American countries over the past few years [18] could be followed by a large dengue epidemic as this temporary cross-protection wanes.

The Pan American Health Organization (PAHO) reported suspected and confirmed Zika cases for every country and territory in the Americas, but these reported cases vastly underestimate the total number of ZIKV infections due to inadequate surveillance, the non-specificity of ZIKV symptoms, and the high proportion of asymptomatic infections [11, 19, 20]. Underreporting is particularly an issue for pathogens such as ZIKV where the majority of infections are asymptomatic or produce only mild symptoms [21, 22]. Estimates of ZIKV infections from blood donors in Puerto Rico through 2016 suggest that almost 470,000 people might have been infected in Puerto Rico alone [23]. High ZIKV seroprevalence estimates from several major cities with populations of more than one million—46% in Managua, Nicaragua [24], 63-68% in Salvador, Brazil [25, 26], and 64% in Recife, Brazil [27]—also suggest that the 806,928 suspected and confirmed cases reported by PAHO represent only a small fraction of the total number of ZIKV infections.

In this study, we take advantage of several features of the recent Zika epidemic that allow us to estimate national and subnational IARs throughout the Americas. First, due to the WHO emergency declaration, surveillance for Zika began in most countries relatively early in the epidemic, and all countries and territories in the Americas reported case data to PAHO. Second, in addition to reporting suspected and confirmed Zika cases in the entire population, many countries also reported Zika cases among pregnant women, CZS or microcephaly cases, and Guillan-Barré syndrome (GBS) cases, providing additional information about the underlying IAR. Third, a number of countries have published subnational Zika surveillance data, which increases the sample sizes for estimation purposes. The subnational data also allows us to capture spatial heterogeneity in ZIKV IAR within these countries. Using a hierarchical Bayesian model fit to multiple data types we estimated the subnational IARs and reporting probabilities for each data type. Estimated IAR and reporting probabilities were then used to extrapolate national-level case data from the rest of the American countries and territories to provide an estimate of the total number of ZIKV infections across this region.

Methods

Data

The cumulative numbers of suspected and confirmed Zika cases in each country and territory in the Americas, as well as the number of confirmed CZS cases, were reported by PAHO on a weekly basis through the first week of 2018 [3]. In addition, PAHO also published country reports following epidemiological week 35 in 2017 that contained additional information, including the total number of cases of microcephaly and Guillan-Barré syndrome (GBS) associated with Zika, where available [28]. Because ZIKV infection attack rates could vary significantly within a country, we restricted our primary analysis to countries and territories where we were able to obtain Zika data at a subnational level for at least one data type. In total, we were able to estimate subnational and national ZIKV IARs for 15 countries and territories (Additional details in S1 Appendix and S1 Table). The included countries were Mexico, all of the countries of Central America (Belize, Costa Rica, El Salvador, Guatemala, Honduras, Nicaragua, Panama), Bolivia, Brazil, Colombia, Ecuador, and Peru in South America, and the Domican Republic and Puerto Rico from the Caribbean. These 15 countries and territories have a combined population of 507.1 million out of a total population of 641.9 million in the Americas (excluding the United States and Canada) and therefore represent a significant fraction of the population at risk.

The data types considered were confirmed Zika cases, suspected Zika cases, microcephaly cases associated with a ZIKV infection in the mother, and Zika-associated cases of GBS. In addition, due to the risk of CZS in newborns, many countries also reported the number of pregnant women with either a suspected or confirmed ZIKV infection, which we treated as seperate data points from the confirmed and suspected cases in the entire population due to the differences in the surveillance and reporting systems for these case types. Suspected Zika cases were defined by WHO/PAHO as a patient with a rash and two or more of the following symptoms: fever, conjunctivitis, arthralgia, myalgia, or peri-articular edema [29]. Reporting of Zika-associated microcephaly and GBS cases varied by country, with some reporting only cases associated with a lab-confirmed ZIKV infection and others both confirmed and suspected cases.

Where available, we obtained Zika data at the first administrative level (e.g., province or state) within a country or territory. Lower level data were aggregated to the first administrative level in cases where they were available. National and subnational population estimates were generated from WorldPop 2015 population rasters for each country or territory [30, 31]. The

number of pregnant women in each country potentially at risk of a ZIKV infection was estimated using 2015 pregnancies rasters from WorldPop, and the number of births at risk for microcephaly were estimated using the WorldPop 2015 births rasters [32]. All of the data used in our analysis, along with model code, are located in the Github repository https://github. com/mooresea/Zika_IAR.

Model

We estimated the national and subnational Zika IAR in each country using a hierarchical Bayesian model with the number of total infections and symptomatic infections treated as latent variables (Fig 1). Short descriptions for each of the model variables and parameters are presented in <u>S2 Table</u>. The model was run separately for each country because a single model would have contained over one thousand parameters, making it computationally prohibitive to simultaneously estimate parameters across all countries. Therefore, posterior parameter estimates for each country are independent of the estimates from the other countries. The IAR in a population of size N_i in administrative unit *i* is the proportion of the population that was infected, $\frac{\mathcal{I}_i}{N}$, where \mathcal{I}_i is the number of infections. The number of symptomatic infections in a population, \mathcal{Z}_i , depends on the size of the infected population, \mathcal{I}_i , and the symptomatic probability, ρ_z , resulting in $\mathcal{Z}_i \sim \text{Bin}(\mathcal{I}_i, \rho_z)$. The number of confirmed Zika cases in the entire population (*T*) of administrative unit *i*, $C_{T,i}$, depends on the number of symptomatic infections and the local reporting probability, $\rho_{C_{T_i}}$, resulting in $C_{T_i} \sim \text{Bin}(\mathcal{Z}_i, \rho_{C_{T_i}})$. The number of suspected Zika cases was similarly dependent on a reporting probability for suspected cases, $\rho_{S_{T_i}}$, and the number of symptomatic infections in the total population, $S_{T_i} \sim Bin(\mathcal{Z}_i, \rho_{S_{T_i}})$. Because misdiagnosis could contribute to the number of suspected cases during an epidemic, we also considered the possibility that there were more suspected cases than symptomatic infections by using a Poisson distribution rather than a binomial distribution to represent the



Fig 1. Model schematic. Model fitting was performed separately for each country or territory. The subscript *i* indicates administrative unit *i* within a modeled country. The subscript *x* represents either the total population (*x* = *T*) or pregnant women only (*x* = *P*). The top row represents the different data types (C = confirmed cases, S = suspected cases, G = Guillan-Barré syndrome cases, M = microcephaly cases). The second row includes the latent variables (Z = Symptomatic infections, and I = Infections). The third row includes the symptompatic probability (ρ_z), the reporting probabilities for confirmed cases ($\rho_{C_{x,i}}$) and suspected cases(ρ_{S_x}), the probability that a symptomatic infection leads to a reported GBS case (ρ_G), and the probability that a ZIKV infection in a pregnant woman leads to a reported microcephaly case (ρ_{M}). The parameters in the bottom row are the hyperparameters for the reporting probabilities $\rho_{C_{x,i}}$ see text and S2 Table for description of model parameters and variables.

https://doi.org/10.1371/journal.pntd.0008640.g001

reporting process for suspected cases. The results of this analysis are reported in the Supporting Information (S3 Appendix). The numbers of confirmed or suspected cases in pregnant women (P), $C_{P,i}$ or $S_{P,i}$, were represented by binomial distributions dependent on the number of symptomatic infections in pregnant women. The infection attack rate and probability of an infection being symptomatic in pregnant women were assumed to be the same as that in the entire population. The probabilities that a symptomatic infection was reported as a suspected or confirmed Zika case, ρ_{S_x} and ρ_{C_x} , where x represents either the entire population (T) or pregnant women only (P), were assumed to differ between administrative units within a country or territory. To estimate this within-country variation in reporting probabilities, we assumed that the probability of a symptomatic infection being reported in administrative unit *i* followed a beta distribution with hyperparameters α_{Y_x} and β_{Y_x} such that $\rho_{Y_{x,i}} \sim \text{Beta}(\alpha_{Y_x}, \beta_{Y_x})$, where Y denotes confirmed (C) or suspected (S).

The number of reported Zika-associated GBS cases was dependent on the number of symptomatic infections, $G_i \sim \text{Bin}(\mathcal{Z}_i, \rho_G)$, where ρ_G is the probability that a symptomatic infection results in a reported GBS case. The number of microcephaly cases associated with Zika was dependent on the total number of births in the population, B_i , and the IAR, such that $M_i \sim \text{Bin}\left(B_i, \rho_M \frac{\mathcal{I}_i}{N_i}\right)$, where ρ_M is the probability that an infection during pregnancy results in a reported microcephaly case.

Subnational IARs were estimated for each country and territory using available data types. The national-level IAR was calculated from the total number of subnational infections divided by the national population size, N, $\sum_{n=1}^{n} T_i$. For Puerto Rico, Zika IARs were estimated at the municipality-level (n = 78), due to availability of data at that scale. In addition, several datasets there were aggregated at the regional level (n = 8), due to their availability at that scale. For these data types, the regional-level IARs were estimated from the total number of infections within all municipalities in a given region.

Prior assumptions. The IAR reported in previous Zika outbreaks has been as high as 73% on the Micronesian island of Yap [21]. The IARs of *Aedes*-transmitted viruses in larger geographical areas tend to be lower than in smaller island environments, such as Yap, because spatial heterogeneity in the presence and abundance of *Aedes* limits transmission potential within a portion of the region [33]. Studies from several different Zika outbreaks have estimated basic reproduction numbers (R_0) of 1.4 to 6.0 [34–38]. Based on the theoretical relationship between R_0 and the final epidemic size [39], these R_0 values would correspond to IARs of 0.286–0.833. However, IARs are typically lower at a given R_0 value in populations with heterogeneous contact patterns [40], as is typical with transmission by *Ae. aegypti* mosquitoes [41]. To lightly constrain our ZIKV IAR estimates without precluding the possibility of values anywhere between 0 and 1, we used a Beta(1, 2) prior for the probability of an individual being infected (i.e., the IAR). This prior distribution had a median value of 0.292 (95% range: 0.013–0.842). We also performed an analysis with a uniform prior for the IARs. A comparison of posterior IAR estimates with and without the Beta(1, 2) prior is included in the Supporting Information (S3 Appendix).

Estimates of the symptomatic probability for Zika, ρ_z , have varied considerably across studies [21, 22, 42]. One recent study estimated the symptomatic probability for three different locations (Yap Island, French Polynesia, and Puerto Rico), taking into account assay sensitivity and specificity, as well as the possibility of Zika-like symptoms due to other causes [22]. Median estimates from that study ranged from 27% in Yap to 50% in Puerto Rico. To generate a single prior distribution for ρ_z in our model, we used the model and data provided in [22] to recreate the posterior estimates of ρ_z from their analysis, and then fitted a beta distribution to the combined posteriors using the 'fitdistrplus' package in R [43]. The resulting distribution was Beta(3.88, 5.34), which has a median of 0.41 and a 95% range of 0.14–0.73, and was used as a prior for each country. The hyperparameters α_{Y_x} and β_{Y_x} for the reporting probabilities ρ_{Y_x} , where Y = C or S and x = T or P, were specified as Cauchy(0, 25) priors. This distribution provides a weakly informative prior with the distribution peaked at 0 and a long right tail. We assumed non-informative priors for the probabilities that a symptomatic infection results in a reported Guillan-Barré case or that an infected mother will give birth to a child with reported microcephaly. These probabilities represent not just the probability that an infection leads to a syndromic case, but also that such a case ends up being reported through the surveillance system.

Model implementation. Each country or territory model was fitted using the 'rstan' version 2.18.2 package in R using a Hamiltonian Monte Carlo algorithm (an MCMC variant) [44]. For each country or territory, four chains of 5,000 iterations each were run with a burn-in interval of 2,500 iterations. Convergence was assessed using the Gelman-Rubin convergence diagnostic [45]. The full model for Peru failed to converge, so the reporting probabilities of confirmed cases (ρ_{C_T}) and confirmed cases in pregnant women (ρ_{C_P}) were estimated as single parameters for all administrative units, rather than drawing ρ_{C_i} from hyperparameters α_C and β_C . We considered this simplification to be justifiable because all ZIKV confirmations were handled by either the national CNS laboratory or one of two regional laboratories, so confirmation rates likely did not vary as widely between administrative units as the reporting probabilities for suspected cases. Model diagnostic results for each country and territory are provided in S2 Appendix.

Model validation

Simulation study. To assess the performance of our estimation method we used the median posterior estimates from Guatemala to simulate new case data for each of the different data types at the first administrative level. Three different simulated datasets were generated with the symptomatic probability (ρ_z) set at either the median posterior estimate from Guatemala (3.2%), 25%, or 50%. We then used our statistical model to estimate the IAR and other parameters either with or without a Beta(2, 1) prior on the IAR parameters.

Posterior predictive checks. Posterior predictive checks were performed by comparing the empirical data to simulated data from the posterior parameter distributions. Posterior predictive data was generated at each iteration, k, of the MCMC, with $Y_j^{pre(k)} \sim Bin(N_j, \theta_j^k)$ for data type $j \in \{C_T, S_T, C_P, S_P, M, G\}$ and its associated parameters θ_j . At each iteration, the observed national total, Y_j , was compared to the predicted national total, $Y_j^{pre(k)}$. This test statistic was used to calculate a Bayesian p-value $p_B = Pr((Y_j^{pre}, \theta_j) \ge (Y_j, \theta_j)|Y_j)$, which indicates whether the distribution of the model-generated data was more extreme than the observed data [45]. We also compared the observed national totals for each data type to the predicted model output to determine whether the observed data (across all case types and territories) fell within the 95% credible interval (CrI) of the corresponding posterior distributions from the model.

Holdout analysis. To determine the sensitivity of model estimates to the inclusion of different data types, we fitted the model while holding out one data type at a time. This analysis was restricted to countries where all data types were available at a subnational level (Guatemala and Dominican Republic) or at most one data type was only available at the national-level (Panama). For these countries, we also fitted the model to one data type at a time to assess the benefit of using multiple data types in the estimation process. **Seroprevalence estimates.** As an additional check, we compared modeled IAR estimates to published seroprevalence estimates from the Americas, as these quantities should be comparable. Seroprevalence estimates from at least one location were available for five countries or territories from ten published studies (S7 Table). Most of these studies used an NS1-based ELISA test for ZIKV IgG antibodies, and several confirmed all postive tests using plaque-reduction neutralization tests (PRNT) or flow cytometry neutralization tests (FRNT). For studies performing both ELISAs and PRNTs, we compared our estimates to the PRNT results. Two included studies did not test for ZIKV IgG: in Guayaquil, Ecuador Zambrano et al. [46] tested a control group of pregnant women for recent ZIKV infection via RT-PCR, and de Araújo [27] tested a control group of new mothers for ZIKV IgM with confirmation via PRNT in Recife, Brazil. The majority of these studies were conducted in a single city and not across a larger adminstrative area, in which case we compared the seroprevalence data to the IAR estimate from the first-level administrative unit where that city was located.

Applying the model to settings with no spatially disaggregated data

Estimates from the 15 country-specific models were used to make predictions about IARs in the 33 other countries and territories in Latin America and the Caribbean where subnational data was not available. The January, 4 2018 Zika report from PAHO included cumulative data from 52 countries and territories in the Americas [3]. Canada, Bermuda, and Chile reported no locally-acquired cases, and the United States reported only a few hundred locally-acquired cases in a limited geographic area, so these countries were excluded from the analysis. For the remaining 48 countries and territories (including the 15 modeled territories), the cumulative numbers of confirmed cases, suspected cases, and cases of microcephaly were used to estimate the national IAR. For each of the 48 countries and territories, a national IAR estimate was obtained by drawing from the posterior distributions of the different reporting parameters from each of the 15 country models. This allowed us to draw from across the full range of estimated reporting probabilities from these 15 countries and territories in predicting the IARs in the remaining countries and territories. For a given country or territory model, k, the probability of a given IAR value in country or territory j was derived from the joint probability for each of the different data types (C_i , S_i , and/or M_i) that were used to fit that model. The combined probability density function for IAR in country *j* was then taken as the sum of the probability density functions using the parameter estimates from all K = 15 models. As an alternative to drawing from all 15 modeled countries and territories to estimate infections in the non-modeled territories, we explored a second method where parameters were drawn only from a subset of the modeled countries that shared a border or similar characteristics (e.g., island nations) with the country or territory being estimated. Additional details on this estimation method are provided in S4 Appendix. These IAR estimates were used to calculate the total number of ZIKV infections that may have occurred during the epidemic. Initially, the infections arising in the 15 modeled territories were derived from country- or territory-specific model estimates and only the infections from the remaining 33 countries and territories were estimated from this pooled analysis. The territory-specific model estimates were compared to the pooled model estimates for each of the 15 modeled territories to assess the plausibility of the pooled estimates in the non-modeled countries and territories (S4 Appendix).

Results

Infection attack rate estimates

Estimated Zika infection attack rates at the national level ranged from 0.084 (95% CrI: 0.067–0.096) in Peru to 0.361 (95% CrI: 0.214–0.514) in Ecuador (Fig 2A, S4 Table). There was





Fig 3. Posterior distribution of subnational ZIKV infection attack rates (IAR) for five different territories (Bolivia, Brazil, Ecuador, Nicaragua, and **Puerto Rico**). Colored circles and whiskers are the median and 95% credible intervals for each administrative unit. Black circles with dashed lines are seroprevalence estimates from the literature (see S7 Table). The dashed lines are the 95% confidence intervals for the seroprevalence estimates assuming a binomial distribution with the exception of the 95% CI estimate from [17] for Bahia, Brazil which was taken directly from their analysis.

https://doi.org/10.1371/journal.pntd.0008640.g003

considerable heterogeneity in IARs (Fig 3). In the most populous country, Brazil, the IAR at the subnational level varied from 0.016 (95% CrI: 0.01-0.025) in the State of Paraná to 0.766 (95% CrI: 0.569-0.942) in the State of Sergipe (Fig 3). In the second most populous country, Mexico, the IAR ranged from 0 (95% CrI: 0-0) in the Federal District to 0.793 (95% CrI: 0.524-0.963) in the State of Yucatán (S17 Fig). Subnational IAR estimates for all 15 modeled countries and territories are presented in S7–S21 Figs.

Mean estimated IARs in countries and territories where subnational data were not available ranged from a low of 0.003 (95% CrI: 0.000-0.013) in Uruguay to a high of 0.979 (95% CrI: 0.591-1.000) in Saint Martin and 0.979 (95% CrI: 0.616-1.000) in Saint Barthélemy (S4 Table, S22–S25 Figs). Summing these IAR estimates across all countries and territories in the Americas, there were a total of 132.3 million (95% CrI: 111.3-170.2 million) ZIKV infections across Latin America and the Caribbean (Table 1). The majority of these infections, 114.1 million (95% CrI: 99.39-128.4 million), were from the 15 modeled countries and territories, while the

	Default Projections	Local Projections
Modeled Countries (N = 15)	114,112,764	114,112,764
(95% CrI)	(99,394,311–128,358,749)	(99,394,311-128,358,749)
Non-modeled Countries (N = 33)	16,160,272	8,766,342
(95% CrI)	(5,426,679–51,709,310)	(4,162,037-35,907,308)
Total Infections	132,278,856	123,996,410
(95% CrI)	(111,305,999–170,156,510)	(108,124,733–151,195,526)

Table 1. Total infections in modeled and projected countries and territories under two different projection methods. Default method used parameter estimates from all 15 modeled countries, while local method used only parameter estimates from neighboring countries and territories or those with similar characteristics.

https://doi.org/10.1371/journal.pntd.0008640.t001

other 33 countries and territories accounted for an additional 16.16 million (95% CrI: 5.427-51.71 million) infections. There were an estimated 53.4 million (95% CrI: 40.8-64.9 million) ZIKV infections in Brazil and 25.6 million (95% CrI: 19.2-32.7 million) in Mexico (Fig 2B). Venezuela had the largest number of ZIKV infections (6.63 million; 95% CrI: 0.35-31.5 million) out of the countries that were not explicitly modeled (S4 Table). The projected number of infections in non-modeled countries was lower under the alternative method of estimation based on only a subset of country-specific parameter estimates (Table 1).

Parameter estimates

The median symptomatic probability of a ZIKV infection across all countries was 0.1 (95% CrI: 0.02-0.53), which was lower than the prior estimate of 0.41 (95% CrI: 0.14–0.73). The estimated symptomatic probability ranged from 0.03 (95% CrI: 0.01-0.17) in Guatemala to 0.33 (95% CrI: 0.15-0.6) in Colombia (Fig 4A). This shift in the posterior estimate relative to the prior estimate of the symptomatic probability may be the result of identifiability issues between the symptomatic probability and the reporting probability parameters for the different data types. Misdiagnosis of symptomatic ZIKV infections as a different disease, such as dengue fever, could also lower the estimate of the symptomatic probability, although dengue virus infections misidentified as ZIKV infections would have the opposite effect.



Fig 4. Posterior parameter estimates. (A) Posterior and prior symptomatic probability estimates for each country or territory. (B) Posterior estimates from each country and territory of the probability that a ZIKV infection in a pregnant woman results in a reported case of microcephaly. Dashed line represents range for estimated risk of Zika-associated microcephaly from published observational studies (see text for references). (C) Posterior estimates from each country and territory of the probability that a symptomatic infection results in a reported Guillan-Barré syndrome (GBS) case.

https://doi.org/10.1371/journal.pntd.0008640.g004

Reported rates of microcephaly in most countries were lower than recent estimates of the risk of microcephaly based on studies of ZIKV infection during pregnancy (Fig 4B). The probability that an infection in a pregnant woman resulted in a reported case of microcephaly ranged from a low of 0.07 per 1,000 infections (95% CrI: 0.01-0.19) in Nicaragua to 8.7 per 1,000 infections (95% CrI: 7.13-11.39) in Brazil. In comparison, estimates of the risk of microcephaly have range from 4.1 (95% CI: 3.4-4.9) per 1,000 [8] to 50 (95% CI: 40-70) per 1,000 births to ZIKV-infected mothers [47]. The probability that a symptomatic infection would result in a reported Guillan-Barré case varied from a low of 0.16 per 100,000 symptomatic infections (95% CrI: 55.72-113.6) in Puerto Rico (Fig 4C). The large variability between countries in reporting probabilities for microcephaly and GBS cases could be due to differences in case definitions among countries, differences in surveillance, or differences in underlying dengue immunity that may have impacted the severity of ZIKV infections due to some form of cross-reactive response [15–17].

The probability of a symptomatic infection being reported as a suspected or confirmed Zika case was higher for pregnant women than the general population for all countries where separate data on pregnant women were available, with the exception of suspected cases in El Salvador and Honduras (S26–S29 Figs). The countrywide reporting probability in pregnant women was as low as 0.002 (95% CrI: 0-0.012) for confirmed cases in El Salvador (S29 Fig, and as high as 0.289 (95% CrI: 0.041-0.989) for confirmed cases in Costa Rica and 0.276 (95% CrI: 0.056-0.727) for suspected cases in Dominican Republic (S28 Fig). As with the variability in reporting probabilities for GBS and microcephaly, this variability in confirming ZIKV infections in pregnant women indicates that there were considerable differences in surveillance and testing efforts among countries during the epidemic.

Countrywide reporting of suspected cases in the entire population ranged from 0.012 (95% CrI: 0.006-0.041) in Peru to 0.933 (95% CrI: 0.565-0.998) in Puerto Rico (S26 Fig). Countrywide reporting of confirmed cases ranged from 0.0004 (95% CrI: 0.0001-0.0023) in El Salvador to 0.504 (95% CrI: 0.305-0.54) in Puerto Rico (S27 Fig). The variation in reporting probabilities among administrative units within a country or territory was largest in Colombia for suspected cases (70.2% of variance was between administrative units vs. 29.8% due to within-unit variance), suspected cases in pregnant women (63.8%), and confirmed cases in pregnant women (67.7%). The largest between-administrative unit variance in reporting probabilities for confirmed cases in the total population occurred in Puerto Rico (56.6% of total variance). Several countries showed little variability in reporting probabilities among administrative units, with < 1% of total variance explained (S30–S33 Figs).

The posterior distribution of the national IAR was only weakly negatively correlated with the symptomatic probability posterior distribution in each country (S34 Fig). The IAR posterior distribution was not strongly correlated with the posterior distributions of any of the reporting probabilities, with the exception of the reporting probability for microcephaly cases, which is the only case type that we assumed could result from both asymptomatic and symptomatic infections (S35–S36 Figs). The posterior distributions of the other reporting rates were strongly correlated with the symptomatic probability posterior distribution (S37 Fig).

Model validation

Simulation study. The modeled posterior IAR estimates closely matched the simulated IAR values when a Beta(1, 2) prior was used, but overestimated IAR when a flat prior was used (S38 Fig). The symptomatic probability (ρ_z) was underestimated for each of the three different simulated values used with or without the Beta prior for IAR (S39 Fig).



Fig 5. Posterior predictive checks at the national level for each data type used in the Bayesian models. Vertical lines are the observed cases and circles are the predicted number of cases with 95% credible intervals for each country and data type.

https://doi.org/10.1371/journal.pntd.0008640.g005

Posterior predictive checks. Our model did not generate predicted cases that were more extreme than the numbers of observed cases—as indicated by Bayesian p-values between 0.1 and 0.9-for each observed data type in all countries and territories except Puerto Rico. For Puerto Rico, the total number of reported confirmed cases ($p_B = 0.990$), suspected cases ($p_B = 0.998$), and GBS cases ($p_B = 0.004$), were outside the range of model predictions, but the number of microcephaly cases was not ($p_B = 0.566$) (Fig 5). The underestimation of confirmed and suspected cases—and overestimation of GBS cases in Puerto Rico indicates that our model was not capable of reconciling estimates based on different data types when the subnational reporting probabilities for these different data types did not vary consistently. Therefore, the municipality-level IAR estimates for Puerto Rico should be viewed with caution. The observed number of microcephaly cases, confirmed cases in pregnant women, and suspected Zika cases in pregnant women fell within the 95% posterior predictive interval (PPI) of the model for every country and territory that reported these data types (Fig 5). Beyond Puerto Rico, the number of confirmed cases in Guatemala was higher than the 95% PPI (1,032 vs. 811-963), the number of suspected cases in Honduras was higher than the 95% PPI (32,385 vs. 29,659-30,583), and the number of suspected cases in Brazil was lower than the 95% PPI (231,725 vs. 240,240-243,040). Comparisons of the observed data to subnational PPIs for each country or territory are provided in the project Github repository https://github.com/mooresea/Zika_IAR.

Holdout analysis. Modeled IAR estimates were not overly sensitive to any one data type. Removing any one data type from the estimation process did not significantly alter the national IAR estimate for Guatemala, with median estimates within 1.0% of the median estimate from the full model of 22.6% (S40 Fig). For the Dominican Republic, the median national IAR estimate varied from 0.217 to 0.314 depending on which data type was withheld, but these median model estimates were all within the 95% CrI for the full model of 0.180-0.331 (S41 Fig). For Panama, the national median estimates with different data types withheld were all slightly higher (0.127-0.150) than the median estimate of the full model (0.117), but still within the 95% CrI of 0.060-0.222 (S42 Fig). When models were fitted with a single data type, national and subnational IAR estimates varied widely depending on which data type was used to fit the model (S43–S45 Figs). The single-data-type models overestimated the median national IAR relative to the full model for all three countries or territories (Guatemala, Panama, and Puerto Rico), and had increased uncertainty at both the national and subnational levels.

Comparison to seroprevalence studies. Our median IAR estimates for Managua, Nicaragua (0.538; 95% CrI: 0.244-0.928) and Guayas, Ecuador (0.503; 95% CrI: 0.153-0.902) were close to seroprevalence estimates from those regions (Managua: 0.46, 95% CI: 0.44-0.48 [24]; Guayas:, 0.50, 95% CI:0.32-0.68 [46]) that were withheld from the fitting process (Fig 3). Overall, the 95% CI for 15 of 18 seroprevalence estimates from published studies overlapped with the 95% CrI of the IAR estimates from our model for the corresponding administrative unit. These two measures are not necessarily identical for every country or territory, because many of the seroprevalence studies were either conducted in only part of the administrative unit for which we estimated the IAR (e.g., the city of Salvador within Bahia State, Brazil or the city of Santos within São Paulo State, Brazil) or were conducted well before the end of the epidemic (Bolivia). The three estimates that did not overlap were Tarija, Bolivia (seroprevalence: 0.005; 95% CI: 0.000-0.015 vs. IAR: 0.108; 95% CrI: 0.043-0.275), Rio Grande do Sul, Brazil (seroprevalence: 0.006; 95% CI: 0.000-0.016 vs. IAR: 0.055; 95% CrI: 0.038-0.078), and Bayamon, Puerto Rico (seroprevalence: 0.347; 95% CI: 0.292-0.412 vs. IAR: 0.536; 95% CrI: 0.475-0.617).

Discussion

Our results show that the vast majority of ZIKV infections in the Americas went unreported during the recent epidemic. PAHO reported fewer than one million suspected or confirmed Zika cases, whereas we estimate that 132.3 million (95% CrI: 111.3-170.2 million) people were infected. This discrepancy is due to the high number of asymptomatic infections, as well as estimated reporting probabilities below 1% for both suspected and confirmed Zika cases in many countries. The majority of ZIKV infections (114.1 million; 95% CrI: 99.39-128.4 million) occurred in the fifteen countries and territories that we modeled subnationally, while an additional 16.16 million (95% CrI: 5.427-51.71 million) occurred in the remaining American territories where national IAR estimates were extrapolated from the estimated reporting probabilities from the modeled countries. These estimates of the national and subnational IAR and total ZIKV infections represent the most extensive attempt to date to estimate the size of the Zika epidemic in the Americas.

Our results show significant within- and between-country variability in IARs. This large variability in IARs could reflect spatial heterogeneity in habitat suitability for *Ae. aegypti* [48]. Early projections of the epidemic also predicted that the IAR would be highly heterogeneous due to the geographic distribution of *Ae. aegypti* as well as other factors likely to influence

transmission intensity, such as temperature and economic prosperity [49]. The historical pattern of virus introduction and spread may have also contributed to the high level of withinand between-country variability in IAR [50]. Local outbreaks in geographic regions or municipalities that began near the end of the local transmission season may have ended before epidemic fadeout would be expected due to herd immunity [10, 51]. Determining the extent to which this spatial heterogeneity is the result of ZIKV epidemiology versus historical contingencies will be important for identifying current and future at-risk populations.

Although our IAR estimates closely matched estimates from seroprevalence studies in some areas (e.g., Managua, Nicaragua, Guayas, Ecuador, and Pernambuco, Brazil), other estimates diverged from published seroprevalence estimates. One potential reason for the discrepancy between our IAR estimates and serosurvey-based estimates is a mismatch in the spatial scale of the region that the IAR estimate covers. For example, our estimate for the Brazilian state of Bahia (0.514; 95% CrI: 0.388-0.631), was slightly lower than the seroprevalence estimates of 0.63 (95% CI: 0.595-0.671),0.68 (95% CI: 0.600-0.744), and 0.73 (95% CI: 0.70-0.76) from Salvador, the largest city in Bahia [17, 25, 26]. However, the overall IAR in Bahia might be lower than the IAR in Salvador if rural areas outside of Salvador experienced a lower infection attack rate. Serosurvey-based estimates might also underestimate IAR if the sensitivity and specificity of the serological assay are not taken into account. Only Rodriguez-Barraquer et al. [17] reported seroprevalence estimates adjusted for assay sensitivity and specificity, which increased their estimate from 0.63 (95% CI: 0.60-0.65) to 0.73 (95% CI: 0.70-0.76) for Salvador, Brazil. Another potential cause of discrepancies is that several of the serosurveys were conducted prior to completion of the ZIKV epidemic. For example, the published seroprevalence estimates for Bolivia from Villarroel et al. [52] were conducted from December, 2016 until May, 2017, but Bolivia had significant ZIKV transmission activity throughout 2017 and 2018. As a result, our IAR estimates which incorporated data from 2017 and 2018 were higher than those reported by Villarroel et al. [52]. Conducting further serosurveys in a range of locations across the Americas would be an important step in determining current levels of immunity and would further validate our model estimates.

Posterior predictive checks showed that Puerto Rico was the only territory for which our model did not accurately predict several of the observed data types, which may explain why our IAR estimate for Bayamon, Puerto Rico was significantly different than the seroprevalence estimate from Lozier et al. [53]. Puerto Rico has more administrative level-one areas (78) than the other modeled countries and territories, which suggests that our model may produce overly precise estimates when there is a large sample size and reporting probabilities are too variable between different administrative levels, confirmed cases were underestimated in some administrative areas of Puerto Rico and overestimated in others. In addition, variation in the reporting probabilities of different case types was not always spatially consistent (e.g., certain areas had above-average reporting of suspected cases, but below average reporting of confirmed cases or GBS cases).

The estimated reporting probabilities for microcephaly in each territory were significantly lower than the reported risk of microcephaly or CZS during pregnancy from several different studies. However, the primary goal of our analysis was to estimate the number of ZIKV infections and not to estimate individual epidemiological parameters such as the symptomatic probability or the risk of Zika-associated microcephaly. Our microcephaly detection estimates are significantly lower than estimates from epidemiological studies because we are estimating both the probability of a ZIKV infection resulting in microcephaly, and the probability that a microcephaly case was diagnosed and reported. Our estimates also depended on estimates of the annual number of births in a region, but this could represent an overestimate of the number of births at risk in many regions because the duration of the epidemic in most localities was less than one year. In addition, differences in the outbreak length among areas would lead to different estimates of the microcephaly reporting probability due to differences in the denominator (number of susceptible births) rather than the underlying risk of microcephaly. Overestimating the number of births at risk would lead to an underestimate of the microcephaly rate in a region or country. Whether this underestimate in the microcephaly rate would also affect the estimate of IAR would depend on the influence of the other data types on the total likelihood. The risk of microcephaly or CZS has been estimated as 0.95% (95% CI: 0.34-1.91%) during the first trimester in French Polynesia [54], 2.3% (2 out of 86 births) in Colombia [55], and 5% (95% CI: 4-7%) among fetuses and infants born to women with laboratoryconfirmed ZIKV infection in the U.S. territories [47]. In a cohort study in Rio de Janerio, 42% of children born to ZIKV-infected mothers were found to have central nervous system abnormalities, although only 4 of 117 were diagnosed with microcephaly [5]. Our estimates ranged from 0.1 per 1,000 infections (95% CrI: 0-0.2) in Nicaragua to 8.7 per 1,000 infections (95% CrI: 7.1-11.4) in Brazil. Changes in the reporting probabilities of both ZIKV infections and microcephaly over the course of the epidemic could also complicate the estimation of the true risk of CZS [56].

Rates of GBS also varied significantly, ranging from 0.04 to 8.51 per 100,000 total ZIKV infections. A recent review estimated a rate of 20 GBS cases per 100,000 ZIKV infections (95% CrI: 5-45 per 100,000), which is higher than our estimates [57]. However, that analysis also estimated the IAR for each location included in their study and estimated lower IARs for Bahia, Brazil (0.02), Salvador, Brazil (0.08), Colombia (0.09), El Salvador (0.15), and Honduras (0.04) than our model. Their IAR estimates for Bahia State and Salvador in Brazil were also much lower than seroprevalence estimates from three studies [17, 25, 26] that more closely matched our IAR estimate of GBS incidence per infection. For Puerto Rico, where our estimate for IAR (0.316; 95% CrI: 0.284–0.347) is within the 95% CrI of the IAR estimate (0.17; 0.08-0.46) by [57], our estimate of 8.51 (95% CrI: 6.94-11.15) GBS cases per 100,000 ZIKV infections. This overlap suggests that our method of simultaneously estimating IAR, reporting probabilities, and the risk of GBS produces reasonable estimates for GBS risk and reporting.

The current status of population-level immunity throughout the Americas has important implications for the future of Zika epidemiology in the region [10, 11, 58]. Our IAR estimates suggest that a number of areas exceeded the herd immunity threshold, which has also been suggested by other studies [19]. However, the high degree of heterogeneity in subnational IARs suggests that certain areas in Central and South America that are suitable for ZIKV transmission may still contain a considerable number of susceptible individuals. While some areas experienced a low IAR because of environmental conditions that limited local transmission, such as high-elevation locations in Colombia and Peru, there are other areas that had a low IAR where conditions appear favorable for ZIKV transmission. For example, the Brazilian states of Pará and Amazonas had median IAR estimates of 0.077 and 0.164, even though both were believed to be high-risk locations [49, 59] based on high suitability for Ae. aegypti [48, 60] and historical dengue incidence [61, 62]. If the remaining population-at-risk is large in these areas, then local outbreaks in the near future could still be possible. Modeling studies also suggest that the presence of multiple locations with insufficient immunity to reach the herd immunity threshold increases the chances that ZIKV becomes endemic in the region in the next decade as new births replenish the susceptible population [10]. Although Zika incidence in the Americas in 2019 appears to be much lower than during the epidemic, limited

transmission has been reported in multiple countries and recent Zika outbreaks have also been reported in Africa and Asia [63], leaving open the possibility that ZIKV could be re-introduced into vulnerable populations. However, it is also possible that there was significant transmission in these states prior to the detection of ZIKV transmission in the Americas, in which case the actual IAR could be higher than we estimate. Genomic data suggests that ZIKV was present in northeast Brazil by early 2014 and spread to multiple countries before Zika surveillance began [64]. Microcephaly data can reconstruct some of this missed portion of the epidemic due to the lag between maternal infection and birth, but even this data will miss infections that occurred more than a year prior to the first case reports.

Because of the high variation in estimated reporting probabilities among modeled countries, there was a high degree of uncertainty in our projections of IAR in the non-modeled territories. Previous studies have also found high variation in the probability of a ZIKV infection being reported between countries [19]. This resulted in a fair amount of uncertainty in our estimates of the total number of ZIKV infections (95% CrI: 111, 305, 999-170, 156, 510) in a total population of 623.7 million. However, because most of the non-modeled territories were relatively small in population size (the total estimated population size in the 15 modeled territories is 507.1 million versus 116.6 million in the 33 non-modeled territories) the median estimate is largely a reflection of infections in the modeled countries. Another limitation of projecting IARs from the modeled countries to non-modeled countries and territories is that smaller islands and overseas territories such as Saint Martin, Guadeloupe, French Guiana, and Martinique likely had better surveillance and higher reporting rates than any of the modeled countries. For example, we projected an attack rate of 1.00 (95% CrI: 0.23-1.00) or 0.59 (95% CrI: 0.41-1.00) in French Guiana, but a recent study by Flamand et al. [65] found an estimated seroprevalence of 0.233 (95% CI: 0.209-0.259), suggesting that their surveillance system detected a higher fraction of infections than most, if not all, of the countries we analyzed. Several overseas territories Because we were unable to precisely estimate IAR without estimating territory-specific reporting probabilities, the publication of subnational totals for different types of Zika cases (suspected, microcephaly, etc.) would help greatly in refining these estimates. The publication of this data in conjunction with seroprevalence data would be particularly valuable for examining differences in surveillance systems and the geographic variation in IAR throughout the entire region [10].

The Zika epidemic in the Americas was a major public health concern that drew considerable attention from both the public health sector and the general public. Despite this attention, the majority of infections were not detected by public health surveillance systems. Using a hierarchical Bayesian model with empirically-informed priors, we were able to leverage Zika case reporting to simultaneously estimate national and subnational reporting probabilities, the fraction of symptomatic infections, and subnational IARs. Our results indicate that fewer than 1% of ZIKV infections in the Americas were reported during the recent epidemic. In the absence of detailed seroprevalence data, our subnational and national IAR estimates are an important first attempt at assessing current levels of ZIKV immunity throughout the Americas. Current levels of immunity are a critical determinant of the probability of further Zika outbreaks—and potentially the likelihood of dengue outbreaks as well—in this region in the next decade.

Supporting information

S1 Appendix. Zika data. (PDF)

S2 Appendix. Model implementation and diagnostics. (PDF)

S3 Appendix. Testing model assumptions. (PDF)

S4 Appendix. Infection attack rate projections. (PDF)

S1 Table. Summary of data availability at the national or subnational level for each country and territory.

(PDF)

S2 Table. Description of each of the variables and parameters used in the models. Parameters were estimated separately for each model country or territory. Where available, the variables were provided at the 1st administrative unit for a particular country. The symptomatic probability (ρ_z), the fraction of (ρ_G) and (ρ_G) were estimated at the country-level. The reporting probabilities for confirmed and suspected cases in the total population and in pregnant women were estimated at the 1st-administrative unit level, using the detailed hyperparameters to capture within-country variation in these reporting probabilities. (PDF)

S3 Table. Comparison of national IAR estimates with or without a beta prior for IAR. (PDF)

S4 Table. Infection attacks rates and total infections for default projections based on parameter estimates from all 15 national models. (PDF)

S5 Table. Infection attacks rates and total infections based on parameter estimates from a subset of national models as indicated by the Models column. (PDF)

S6 Table. Comparison of model-specific IAR estimates for each modeled territory versus projections of IAR from combined model estimates under the two different projection methods.

(PDF)

S7 Table. Zika seroprevalence studies. (PDF)

S1 Fig. Traceplots of the log probability density of the posterior distribution for each country model after warmup. Colors represent the four separate chains. (TIF)

S2 Fig. Comparison of posterior national-level IAR estimates for each country or territory with or without a Beta(1, 2) prior for subnational IARs. (TIF)

S3 Fig. Comparison of posterior estimates from four countries of the probability that a symptomatic infection was reported as a suspected case when the reporting process was modeled using a binomial or Poisson distribution. (TIF)

S4 Fig. Comparison of posterior national-level IAR estimates for four countries when the reporting process for suspected cases was modeled using a binomial or Poisson

distribution. (TIF)

S5 Fig. Comparison of posterior national-level IAR estimates for four countries when the reporting process for suspected cases was modeled using a binomial or Poisson distribution. (TIF)

S6 Fig. Comparison of posterior national-level IAR estimates for four countries when the reporting process for suspected cases was modeled using a binomial or Poisson distribution.

(TIF)

S7 Fig. Posterior distribution of subnational infection attack rates (IAR) for Belize. (TIF)

S8 Fig. Posterior distribution of subnational infection attack rates (IAR) for Bolivia. (TIF)

S9 Fig. Posterior distribution of subnational infection attack rates (IAR) for Brazil. (TIF)

S10 Fig. Posterior distribution of subnational infection attack rates (IAR) for Colombia. (TIF)

S11 Fig. Posterior distribution of subnational infection attack rates (IAR) for Costa Rica. (TIF)

S12 Fig. Posterior distribution of subnational infection attack rates (IAR) for the Dominican republic. (TIF)

S13 Fig. Posterior distribution of subnational infection attack rates (IAR) for Ecuador. (TIF)

S14 Fig. Posterior distribution of subnational infection attack rates (IAR) for El Salvador. (TIF)

S15 Fig. Posterior distribution of subnational infection attack rates (IAR) for Guatemala. (TIF)

S16 Fig. Posterior distribution of subnational infection attack rates (IAR) for Honduras. (TIF)

S17 Fig. Posterior distribution of subnational infection attack rates (IAR) for Mexico. (TIF)

S18 Fig. Posterior distribution of subnational infection attack rates (IAR) for Nicaragua. (TIF)

S19 Fig. Posterior distribution of subnational infection attack rates (IAR) for Panama. (TIF)

S20 Fig. Posterior distribution of subnational infection attack rates (IAR) for Peru. (TIF)

S21 Fig. Posterior distribution of subnational infection attack rates (IAR) for Puerto Rico. (TIF)

S22 Fig. Probability distribution of the infection attack rate (IAR) for the Bahamas assuming reporting probabilities estimated from (A) all 15 different national models or (B) a subset of national models based on similarities in the country or territories. The different colors represent the probability distribution of IAR generated from using the estimated reporting probabilities from each modeled territory. (TIF)

S23 Fig. Probability distribution of the infection attack rate (IAR) for Venezuela assuming reporting probabilities estimated from (A) all 15 different national models or (B) a subset of national models based on similarities in the country or territories. The different colors represent the probability distribution of IAR generated from using the estimated reporting probabilities from each modeled territory. (TIF)

S24 Fig. Probability distribution of the infection attack rate (IAR) for the British Virgin Islands assuming reporting probabilities estimated from (A) all 15 different national models or (B) a subset of national models based on similarities in the country or territories. The different colors represent the probability distribution of IAR generated from using the estimated reporting probabilities from each modeled territory. (TIF)

S25 Fig. Probability distribution of the infection attack rate (IAR) for the U.S. Virgin Islands assuming reporting probabilities estimated from (A) all 15 different national models or (B) a subset of national models based on similarities in the country or territories. The different colors represent the probability distribution of IAR generated from using the estimated reporting probabilities from each modeled territory. (TIF)

S26 Fig. Posterior estimates from each country and territory of the probability that a symptomatic ZIKV infection is reported as a suspected Zika case. (TIF)

S27 Fig. Posterior estimates from each country and territory of the probability that a symptomatic ZIKV infection is reported as a confirmed Zika case. (TIF)

S28 Fig. Posterior estimates from each country and territory of the probability that a symptomatic ZIKV infection in a pregnant woman is reported as a suspected Zika case. (TIF)

S29 Fig. Posterior estimates from each country and territory of the probability that a symptomatic ZIKV infection in a pregnant woman is reported as a confirmed Zika case. (TIF)

S30 Fig. Posterior estimates from each country and territory of the probability that a symptomatic ZIKV infection is reported as a suspected Zika case. Black lines show the country-wide average reporting probability and grey lines show the estimated reporting probability in each administrative unit. (TIF)

IIF)

S31 Fig. Posterior estimates from each country and territory of the probability that a symptomatic ZIKV infection is reported as a confirmed Zika case. Black lines show the country-wide average reporting probability and grey lines show the estimated reporting

probability in each administrative unit. (TIF)

S32 Fig. Posterior estimates from each country and territory of the probability that a symptomatic ZIKV infection in a pregnant woman is reported as a suspected Zika case. Black lines show the country-wide average reporting probability and grey lines show the estimated reporting probability in each administrative unit. (TIF)

S33 Fig. Posterior estimates from each country and territory of the probability that a symptomatic ZIKV infection in a pregnant woman is reported as a confirmed Zika case. Black lines show the country-wide average reporting probability and grey lines show the estimated reporting probability in each administrative unit. (TIF)

S34 Fig. Joint posterior distributions for infection attack rate (IAR) and symptomatic probability from each country and territory. (TIF)

S35 Fig. Joint posterior distributions for infection attack rate (IAR) and different reporting probabilities for Colombia. (TIF)

S36 Fig. Joint posterior distributions for infection attack rate (IAR) and microcephaly reporting probability from each country and territory. (TIF)

S37 Fig. Joint posterior distributions for symptomatic probability and different reporting probabilities for Colombia. (TIF)

S38 Fig. Posterior distributions of ZIKV infection attack rates (IAR) estimated from simulated data. Simulation numbers are the three different simulated datasets with different symptomatic probability values. Blue values are posterior distributions from model using a Beta(1, 2) prior, and red values are posterior distributions from model with a flat prior. The solid lines are the simulated IAR value being estimated. (TIF)

S39 Fig. Posterior distributions of the symptomatic probability (ρ_z) estimated from simulated data. Simulation numbers are the three different simulated datasets with different ρ_z values. Blue values are posterior distributions from model using a Beta(1, 2) prior, and red values are posterior distributions from model with a flat prior. The solid lines are the simulated ρ_z value being estimated.

(TIF)

S40 Fig. Posterior distributions of national and subnational ZIKV infection attack rates (IAR) for Guatemala when each of the different data types is withheld from the model fitting process.

(TIF)

S41 Fig. Posterior distributions of national and subnational ZIKV infection attack rates (IAR) for the Dominican Republic when each of the different data types is withheld from the model fitting process.

(TIF)

S42 Fig. Posterior distributions of national and subnational ZIKV infection attack rates (IAR) for Panama when each of the different data types is withheld from the model fitting process.

(TIF)

S43 Fig. Posterior distributions of national and subnational ZIKV infection attack rates (IAR) for Guatemala from models fit to a single data type in comparison to the full model fit to all data types.

(TIF)

S44 Fig. Posterior distributions of national and subnational ZIKV infection attack rates (IAR) for the Dominican Republic from models fit to a single data type in comparison to the full model fit to all data types. (TIF)

S45 Fig. Posterior distributions of national and subnational ZIKV infection attack rates (IAR) for Panama from models fit to a single data type in comparison to the full model fit to all data types.

(TIF)

Acknowledgments

We would like to thank the Epidemiology Unit of the Belize Ministry of Health for providing Zika reports upon request. We also thank all of the governments in North and South America that made their Zika surveillance reports or data publicly available online.

Author Contributions

Conceptualization: Sean M. Moore, Robert C. Reiner, Jr., Christopher M. Barker, T. Alex Perkins.

Data curation: Sean M. Moore, Rachel J. Oidtman, K. James Soda, Amir S. Siraj.

Formal analysis: Sean M. Moore, T. Alex Perkins.

Methodology: Sean M. Moore, Robert C. Reiner, Jr., Christopher M. Barker, T. Alex Perkins.

Writing – review & editing: Sean M. Moore, Rachel J. Oidtman, K. James Soda, Amir S. Siraj, Robert C. Reiner, Jr., Christopher M. Barker, T. Alex Perkins.

References

- Dick GWA, Kitchen SF, Haddow AJ. Zika Virus (I). Isolations and serological specificity. Transactions of The Royal Society of Tropical Medicine and Hygiene. 1952; 46(5):509–520. <u>https://doi.org/10.1016/</u> 0035-9203(52)90042-4 PMID: 12995440
- 2. Fauci AS, Morens DM. Zika Virus in the Americas: Yet Another Arbovirus Threat. New England Journal of Medicine. 2016; 374(7):601–604. https://doi.org/10.1056/NEJMp1600297 PMID: 26761185
- Pan American Health Organization (PAHO). Zika cases and congenital syndrome associated with Zika virus reported by countries and territories in the Americas, 2015-2018 cumulative cases; 2018. Available from: https://www.paho.org/hq/index.php?option=com_content&view=article&id=12390:zika-cumulative-cases&Itemid=42090&Iang=en.
- WHO. WHO Statement on the First Meeting of the International Health Regulations (2005) (IHR 2005) Emergency Committee on Zika Virus and Observed Increase in Neurological Disorders and Neonatal Malformations. WHO; 2016. Available from: http://www.who.int/mediacentre/news/statements/2016/ 1st-emergency-committee-zika/en/.

- Brasil P, Pereira JP, Moreira ME, Ribeiro Nogueira RM, Damasceno L, Wakimoto M, et al. Zika Virus Infection in Pregnant Women in Rio de Janeiro. New England Journal of Medicine. 2016; 375 (24):2321–2334. https://doi.org/10.1056/NEJMoa1602412 PMID: 26943629
- Mlakar J, Korva M, Tul N, Popović M, Poljšak-Prijatelj M, Mraz J, et al. Zika Virus Associated with Microcephaly. New England Journal of Medicine. 2016; 374(10):951–958. https://doi.org/10.1056/ NEJMoa1600651 PMID: 26862926
- Rasmussen SA, Jamieson DJ, Honein MA, Petersen LR. Zika Virus and Birth Defects–Reviewing the Evidence for Causality. The New England Journal of Medicine. 2016; 374(20):1981–1987. https://doi. org/10.1056/NEJMsr1604338 PMID: 27074377
- Brady OJ, Osgood-Zimmerman A, Kassebaum NJ, Ray SE, Araújo VEMd, Nóbrega AAd, et al. The association between Zika virus infection and microcephaly in Brazil 2015-2017: An observational analysis of over 4 million births. PLOS Medicine. 2019; 16(3):e1002755. https://doi.org/10.1371/journal. pmed.1002755 PMID: 30835728
- Broutet N, Krauer F, Riesen M, Khalakdina A, Almiron M, Aldighieri S, et al. Zika Virus as a Cause of Neurologic Disorders. New England Journal of Medicine. 2016; 374(16):1506–1509. https://doi.org/10. 1056/NEJMp1602708 PMID: 26959308
- Ferguson NM, Cucunubá ZM, Dorigatti I, Nedjati-Gilani GL, Donnelly CA, Basáñez MG, et al. Countering the Zika epidemic in Latin America. Science. 2016; 353(6297):353–354. <u>https://doi.org/10.1126/</u> science.aag0219 PMID: 27417493
- Lessler J, Chaisson LH, Kucirka LM, Bi Q, Grantz K, Salje H, et al. Assessing the global threat from Zika virus. Science. 2016; 353(6300):aaf8160. https://doi.org/10.1126/science.aaf8160 PMID: 27417495
- Durham DP, Fitzpatrick MC, Ndeffo-Mbah ML, Parpia AS, Michael NL, Galvani AP. Evaluating vaccination strategies for Zika virus in the Americas. Annals of Internal Medicine. 2018; 168(9):621–630. https://doi.org/10.7326/M17-0641 PMID: 29610863
- Stettler K, Beltramello M, Espinosa DA, Graham V, Cassotta A, Bianchi S, et al. Specificity, cross-reactivity, and function of antibodies elicited by Zika virus infection. Science. 2016; 353(6301):823–826. https://doi.org/10.1126/science.aaf8505 PMID: 27417494
- Dejnirattisai W, Supasa P, Wongwiwat W, Rouvinski A, Barba-Spaeth G, Duangchinda T, et al. Dengue virus sero-cross-reactivity drives antibody-dependent enhancement of infection with zika virus. Nature Immunology. 2016; 17(9):1102–1108. https://doi.org/10.1038/ni.3515 PMID: 27339099
- Bardina SV, Bunduc P, Tripathi S, Duehr J, Frere JJ, Brown JA, et al. Enhancement of Zika virus pathogenesis by preexisting antiflavivirus immunity. Science. 2017; 356(6334):175–180. <u>https://doi.org/10.1126/science.aal4365</u> PMID: 28360135
- Andrade DV, Harris E. Recent advances in understanding the adaptive immune response to Zika virus and the effect of previous flavivirus exposure. Virus Research. 2018; 254:27–33. https://doi.org/10. 1016/j.virusres.2017.06.019 PMID: 28655548
- Rodriguez-Barraquer I, Costa F, Nascimento EJM, Nery N, Castanha PMS, Sacramento GA, et al. Impact of preexisting dengue immunity on Zika virus emergence in a dengue endemic region. Science. 2019; 363(6427):607–610. https://doi.org/10.1126/science.aav6618 PMID: 30733412
- Borchering RK, Huang AT, Mier-y Teran-Romero L, Rojas DP, Rodriguez-Barraquer I, Katzelnick LC, et al. Impacts of Zika emergence in Latin America on endemic dengue transmission. Nature Communications. 2019; 10(1):5730. https://doi.org/10.1038/s41467-019-13628-x PMID: 31844054
- O'Reilly KM, Lowe R, Edmunds WJ, Mayaud P, Kucharski A, Eggo RM, et al. Projecting the end of the Zika virus epidemic in Latin America: a modelling analysis. BMC Medicine. 2018; 16(1):180. https://doi. org/10.1186/s12916-018-1158-8 PMID: 30285863
- Burger-Calderon R, Bustos Carrillo F, Gresh L, Ojeda S, Sanchez N, Plazaola M, et al. Age-dependent manifestations and case definitions of paediatric Zika: a prospective cohort study. The Lancet Infectious Diseases. 2020; 20(3):371–380. https://doi.org/10.1016/S1473-3099(19)30547-X PMID: 31870907
- Duffy MR, Chen TH, Hancock WT, Powers AM, Kool JL, Lanciotti RS, et al. Zika Virus Outbreak on Yap Island, Federated States of Micronesia. New England Journal of Medicine. 2009; 360(24):2536–2543. https://doi.org/10.1056/NEJMoa0805715 PMID: 19516034
- Mitchell PK, Mier-y Teran-Romero L, Biggerstaff BJ, Delorey MJ, Aubry M, Cao-Lormeau VM, et al. Reassessing Serosurvey-Based Estimates of the Symptomatic Proportion of Zika Virus Infections. American Journal of Epidemiology. 2019; 188(1):206–213. <u>https://doi.org/10.1093/aje/kwy189</u> PMID: 30165474
- Chevalier MS, Biggerstaff BJ, Basavaraju SV, Ocfemia MCB, Alsina JO, Climent-Peris C, et al. Use of Blood Donor Screening Data to Estimate Zika Virus Incidence, Puerto Rico, April-August 2016. Emerging Infectious Diseases. 2017; 23(5):790–795. <u>https://doi.org/10.3201/eid2305.161873</u> PMID: 28263141

- Zambrana JV, Carrillo FB, Burger-Calderon R, Collado D, Sanchez N, Ojeda S, et al. Seroprevalence, risk factor, and spatial analyses of Zika virus infection after the 2016 epidemic in Managua, Nicaragua. Proceedings of the National Academy of Sciences. 2018; 115(37):9294–9299. https://doi.org/10.1073/ pnas.1804672115 PMID: 30150394
- Netto EM, Moreira-Soto A, Pedroso C, Höser C, Funk S, Kucharski AJ, et al. High Zika Virus Seroprevalence in Salvador, Northeastern Brazil Limits the Potential for Further Outbreaks. mBio. 2017; 8(6): e01390–17. https://doi.org/10.1128/mBio.01390-17 PMID: 29138300
- Moreira-Soto A, Cabral R, Pedroso C, Eschbach-Bludau M, Rockstroh A, Vargas LA, et al. Exhaustive TORCH Pathogen Diagnostics Corroborate Zika Virus Etiology of Congenital Malformations in Northeastern Brazil. mSphere. 2018; 3(4):e00278–18. https://doi.org/10.1128/mSphere.00278-18 PMID: 30089647
- 27. de Araújo TVB, Rodrigues LC, de Alencar Ximenes RA, de Barros Miranda-Filho D, Montarroyos UR, de Melo APL, et al. Association between Zika virus infection and microcephaly in Brazil, January to May, 2016: preliminary report of a case-control study. The Lancet Infectious Diseases. 2016; 16 (12):1356–1363. https://doi.org/10.1016/S1473-3099(16)30318-8 PMID: 27641777
- Pan American Health Organization (PAHO); 2017. Available from: https://www.paho.org/hq/index.php? option=com_content&view=article&id=11603:countries-and-territories-with-autochthonoustransmission-of-zika-virus-in-the-americas-reported-in-2015-2017&Itemid=41696&Iang=en.
- 29. Pan American Health Organization (PAHO). Zika Resources: Case Definitions; 2016. Available from: https://www.paho.org/hq/index.php?option=com_content&view=article&id=11117:zika-resourcescase-definitions&Itemid=41532&Iang=en.
- 30. WorldPop; 2019. Available from: www.worldpop.org.
- Sorichetta A, Hornby GM, Stevens FR, Gaughan AE, Linard C, Tatem AJ. High-resolution gridded population datasets for Latin America and the Caribbean in 2010, 2015, and 2020. Scientific Data. 2015; 2:150045. https://doi.org/10.1038/sdata.2015.45 PMID: 26347245
- Tatem AJ, Campbell J, Guerra-Arias M, de Bernis L, Moran A, Matthews Z. Mapping for maternal and newborn health: the distributions of women of childbearing age, pregnancies and births. International Journal of Health Geographics. 2014; 13:2. https://doi.org/10.1186/1476-072X-13-2 PMID: 24387010
- Keeling MJ. The effects of local spatial structure on epidemiological invasions. Proceedings of the Royal Society B: Biological Sciences. 1999; 266(1421):859–867. <u>https://doi.org/10.1098/rspb.1999</u>. 0716 PMID: 10343409
- Kucharski AJ, Funk S, Eggo RM, Mallet HP, Edmunds WJ, Nilles EJ. Transmission Dynamics of Zika Virus in Island Populations: A Modelling Analysis of the 2013-14 French Polynesia Outbreak. PLOS Neglected Tropical Diseases. 2016; 10(5):e0004726. https://doi.org/10.1371/journal.pntd.0004726 PMID: 27186984
- **35.** Towers S, Brauer F, Castillo-Chavez C, Falconar AKI, Mubayi A, Romero-Vivas CME. Estimate of the reproduction number of the 2015 Zika virus outbreak in Barranquilla, Colombia, and estimation of the relative role of sexual transmission. Epidemics. 2016; 17:50–55. https://doi.org/10.1016/j.epidem.2016. 10.003 PMID: 27846442
- Lourenço J, Maia de Lima M, Faria NR, Walker A, Kraemer MU, Villabona-Arenas CJ, et al. Epidemiological and ecological determinants of Zika virus transmission in an urban setting. eLife. 2017; 6: e29820. https://doi.org/10.7554/eLife.29820 PMID: 28887877
- Shutt DP, Manore CA, Pankavich S, Porter AT, Del Valle SY. Estimating the reproductive number, total outbreak size, and reporting rates for Zika epidemics in South and Central America. Epidemics. 2017; 21:63–79. https://doi.org/10.1016/j.epidem.2017.06.005 PMID: 28803069
- Villela DaM, Bastos LS, Carvalho LMD, Cruz OG, Gomes MFC, Durovni B, et al. Zika in Rio de Janeiro: Assessment of basic reproduction number and comparison with dengue outbreaks. Epidemiology & Infection. 2017; 145(8):1649–1657. https://doi.org/10.1017/S0950268817000358 PMID: 28240195
- **39.** Keeling MJ, Rohani P. Modeling Infectious Diseases in Humans and Animals. Princeton University Press; 2011.
- Andreasen V. The final size of an epidemic and its relation to the basic reproduction number. Bulletin of Mathematical Biology. 2011; 73(10):2305–2321. https://doi.org/10.1007/s11538-010-9623-3 PMID: 21210241
- Liebman KA, Stoddard ST, Jr RCR, Perkins TA, Astete H, Sihuincha M, et al. Determinants of Heterogeneous Blood Feeding Patterns by Aedes aegypti in Iquitos, Peru. PLOS Neglected Tropical Diseases. 2014; 8(2):e2702. https://doi.org/10.1371/journal.pntd.0002702 PMID: 24551262
- Subissi L, Daudens-Vaysse E, Cassadou S, Ledrans M, Bompard P, Gustave J, et al. Revising rates of asymptomatic Zika virus infection based on sentinel surveillance data from French Overseas Territories. International Journal of Infectious Diseases. 2017; 65:116–118. https://doi.org/10.1016/j.ijid.2017. 10.009 PMID: 29081369

- Delignette-Muller ML, Dutang C. fitdistrplus: An R Package for Fitting Distributions. Journal of Statistical Software. 2015; 64(4):1–34. https://doi.org/10.18637/jss.v064.i04
- 44. Stan Development Team. RStan: the R interface to Stan; 2018. Available from: http://mc-stan.org/.
- 45. Gelman A, Carlin JB, Stern HS, Dunson DB, Vehtari A, Rubin DB. Bayesian Data Analysis. 3rd ed. Boca Raton: Chapman and Hall/CRC; 2013.
- 46. Zambrano H, Waggoner J, León K, Pinsky B, Vera K, Schettino M, et al. High incidence of Zika virus infection detected in plasma and cervical cytology specimens from pregnant women in Guayaquil, Ecuador. American Journal of Reproductive Immunology. 2017; 77(2):e12630. https://doi.org/10.1111/ aji.12630 PMID: 28177195
- Shapiro-Mendoza CK. Pregnancy Outcomes After Maternal Zika Virus Infection During Pregnancy—U. S. Territories, January 1, 2016-April 25, 2017. MMWR Morbidity and Mortality Weekly Report. 2017; 66. https://doi.org/10.15585/mmwr.mm6623e1 PMID: 28617773
- Kraemer MU, Sinka ME, Duda KA, Mylne AQ, Shearer FM, Barker CM, et al. The global distribution of the arbovirus vectors Aedes aegypti and Ae. albopictus. eLife. 2015; 4:e08347. https://doi.org/10.7554/ eLife.08347 PMID: 26126267
- Perkins TA, Siraj AS, Ruktanonchai CW, Kraemer MUG, Tatem AJ. Model-based projections of Zika virus infections in childbearing women in the Americas. Nature Microbiology. 2016; 1:16126. <u>https://doi.org/10.1038/nmicrobiol.2016.126 PMID</u>; 27562260.
- Zhang Q, Sun K, Chinazzi M, Piontti APy, Dean NE, Rojas DP, et al. Spread of Zika virus in the Americas. Proceedings of the National Academy of Sciences. 2017; 114(22):E4334–E4343. https://doi.org/ 10.1073/pnas.1620161114 PMID: 28442561
- Huber JH, Childs ML, Caldwell JM, Mordecai EA. Seasonal temperature variation influences climate suitability for dengue, chikungunya, and Zika transmission. PLOS Neglected Tropical Diseases. 2018; 12(5):e0006451. https://doi.org/10.1371/journal.pntd.0006451 PMID: 29746468
- Villarroel PMS, Nurtop E, Pastorino B, Roca Y, Drexler JF, Gallian P, et al. Zika virus epidemiology in Bolivia: A seroprevalence study in volunteer blood donors. PLOS Neglected Tropical Diseases. 2018; 12(3):e0006239. https://doi.org/10.1371/journal.pntd.0006239 PMID: 29513667
- Lozier MJ, Burke RM, Lopez J, Acevedo V, Amador M, Read JS, et al. Differences in Prevalence of Symptomatic Zika Virus infection, by Age and Sex-Puerto Rico, 2016. The Journal of Infectious Diseases. 2018; 217(11):1678–1689. https://doi.org/10.1093/infdis/jix630 PMID: 29216376
- Cauchemez S, Besnard M, Bompard P, Dub T, Guillemette-Artur P, Eyrolle-Guignot D, et al. Association between Zika virus and microcephaly in French Polynesia, 2013-15: a retrospective study. The Lancet. 2016; 387(10033):2125–2132. https://doi.org/10.1016/S0140-6736(16)00651-6 PMID: 26993883
- 55. Rodriguez-Morales AJ, Cardona-Ospina JA, Ramirez-Jaramillo V, Gaviria JA, González-Moreno GM, Castrillón-Spitia JD, et al. Diagnosis and outcomes of pregnant women with Zika virus infection in two municipalities of Risaralda, Colombia: Second report of the ZIKERNCOL study. Travel Medicine and Infectious Disease. 2018; 25:20–25. https://doi.org/10.1016/j.tmaid.2018.06.006 PMID: 29894797
- Hay JA, Nouvellet P, Donnelly CA, Riley S. Potential inconsistencies in Zika surveillance data and our understanding of risk during pregnancy. PLOS Neglected Tropical Diseases. 2018; 12(12):e0006991. https://doi.org/10.1371/journal.pntd.0006991 PMID: 30532143
- Mier-y Teran-Romero L, Delorey MJ, Sejvar JJ, Johansson MA. Guillain-Barré syndrome risk among individuals infected with Zika virus: a multi-country assessment. BMC Medicine. 2018; 16(1):67. <u>https://</u> doi.org/10.1186/s12916-018-1052-4 PMID: 29759069
- Perkins TA. Retracing Zika's footsteps across the Americas with computational modeling. Proceedings of the National Academy of Sciences. 2017; 114(22):5558–5560. <u>https://doi.org/10.1073/pnas.</u> 1705969114 PMID: 28533416
- Messina JP, Kraemer MU, Brady OJ, Pigott DM, Shearer FM, Weiss DJ, et al. Mapping global environmental suitability for Zika virus. eLife. 2016; 5:e15272. <u>https://doi.org/10.7554/eLife.15272</u> PMID: 27090089
- Brady OJ, Golding N, Pigott DM, Kraemer MUG, Messina JP, Reiner RC Jr, et al. Global temperature constraints on Aedes aegypti and Ae. albopictus persistence and competence for dengue virus transmission. Parasites & Vectors. 2014; 7(1):338. <u>https://doi.org/10.1186/1756-3305-7-338</u> PMID: 25052008
- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, Moyes CL, et al. The global distribution and burden of dengue. Nature. 2013; 496(7446):504–507. https://doi.org/10.1038/nature12060 PMID: 23563266
- Siqueira JaB, Martelli CMT, Coelho GE, Simplício ACdR, Hatch DL. Dengue and Dengue Hemorrhagic Fever, Brazil, 1981-2002. Emerging Infectious Diseases. 2005; 11(1):48–53. <u>https://doi.org/10.3201/</u> eid1101.031091 PMID: 15705322

- 63. ProMED-mail. Zika virus (02): Americas, Asia, research, observations. ProMED-mail. 2019;Archive Number: 20190320.63777153.
- 64. Faria NR, Quick J, Claro IM, Thézé J, de Jesus JG, Giovanetti M, et al. Establishment and cryptic transmission of Zika virus in Brazil and the Americas. Nature. 2017; 546(7658):406–410. https://doi.org/10. 1038/nature22401 PMID: 28538727
- Flamand C, Bailly S, Fritzell C, Berthelot L, Vanhomwegen J, Salje H, et al. Impact of Zika Virus Emergence in French Guiana: A Large General Population Seroprevalence Survey. The Journal of Infectious Diseases. 2019; 220(12):1915–1925. https://doi.org/10.1093/infdis/jiz396 PMID: 31418012

PLOS Neglected Tropical Diseases | https://doi.org/10.1371/journal.pntd.0008640 September 28, 2020